Author's response to reviews

Title: Time delays in diagnosis of pulmonary tuberculosis: a systematic review of literature

Authors:

Chandrashekhar T Sreeramareddy (chandrashekharats@yahoo.com)
Panduru V Kishore (pandu_vk@yahoo.com)
Joris Menten (imenten@itg.be)
J Van den Ende (jvdende@itg.be)

Version: 4 Date: 4 June 2009

Author's response to reviews:

Dear Editorial Production Team,

Thank you for e-mail with details of changes/formatting changes to made to the manuscript.

Please find below the details of changes that were made:
1. Abstract in the manuscript submitted in to the submission system
2. Suggested changes were made to the references
3. Title was made into the sentence case
4. Introduction renamed as background
5. Authors contributions formatted into a paragraph
6. Sources of funding shifted to acknowledgements
7. Table vertical elements were tried to make a relayout but disturbed the whole table so it has been retained.
8. Also we could not reformat the figures i.e. cropping up

Hope that the changes made are acceptable

Regards

Chandrashekhar